Author’s response to reviews

Title: Development of estimates of dietary nitrates, nitrites, and nitrosamines for use with the Short Willet Food Frequency Questionnaire

Authors:

John S Griesenbeck (jsgriesenbeck@srph.tamhsc.edu)
Michelle D Steck (mdsteck@srph.tamhsc.edu)
John C Huber Jr (jchuber@srph.tamhsc.edu)
Joseph R Sharkey (jrsharkey@srph.tamhsc.edu)
Antonio A Rene (aarene@srph.tamhsc.edu)
Jean D Brender (jdbrender@srph.tamhsc.edu)

Version: 5 Date: 31 March 2009

Author’s response to reviews:

We have responded to the comments from the Editorial Production Team.